Author's response to reviews

Title: Event-based record linkage in health and aged care services data: a methodological innovation

Authors:

Rosemary Karmel (rosemary.karmel@aihw.gov.au)
Diane Gibson (diane.gibson@aihw.gov.au)

Version: 5 Date: 20 September 2007

Author's response to reviews:

I have addressed the requested formatting changes as follows:
1. reference ranges rather than lists have been used where appropriate
2. table and figure placement text has been removed
3. 'Abbreviations' title has been corrected.
4. hard return separating authors' contributions has been deleted, as have been additional hard returns between paragraphs
5. Source in table legend and figure legend has been moved to reference list
6. additional files section has been deleted (as unnecessary)
7. figure files have been amended to delete figure titles and legends
8. ENDNOTE references have been corrected so that there is correspondence between references in the manuscript and those extracted by ENDNOTE in the reference list.
9. Minor grammatical corrections have been made to the manuscript

Regards

Rosemary Karmel